Author’s response to reviews

Title: Immunohistochemical staining of leptin is associated with grade, stage, lymph node involvement, recurrence, and hormone receptor phenotypes in breast cancer

Authors:
Mohamad Nidal Khabaz (mnkhabaz@kau.edu.sa)
Amer Abdelrahman (aaaali@kau.edu.sa)
Nadeem Butt (nshafique@kau.edu.sa)
Lila Damnhory (lailahhd71@hotmail.com)
Mohamed Elshal (melshal2002@yahoo.com)
Alia Al-Dahlawi (alia008@hotmail.com)
Swsan Ashoor (sawsan_ashoor@hotmail.com)
Basim Al-Maghrabi (Bjalmaghrabi@gmail.com)
Pauline Dobson (pauladobro1911@gmail.com)
Barry Brown (barleybro41@outlook.com)
Kaltoom Al-Sakkaf (kalsakkaf@kau.edu.sa)
Mohmmad Al-Qahtani (mhalqahtani@kau.edu.sa)
Jaudah Al-Maghrabi (jalmgrabi@kau.edu.sa)

Version: 2 Date: 20 Sep 2017

Author’s response to reviews:

Response to reviewer

We greatly appreciate the efforts of reviewers and editorial staff for their efforts in the review process of our manuscript.

The revised manuscript conforms to the journal style.
We note that the current submission contains some textual overlap with other previously published works. Please refer to the attached document and re-phrase the highlighted sections in your own words to ensure no overlap.

On behalf of myself and all the co-authors we do apologize for the overlap that occurred in this manuscript with the other previously published papers. All the overlapped sections have been re-phrased and we ensure that there is no overlap.

Additionally, in your cover letter please explain how these two papers are different, and what they separately contribute to the scientific understanding of this field.

Explanation has been mention in our cover letter.